

Citation: Saad-Roy CM, Morris SE, Boots M, Baker RE, Lewis BL, Farrar J, et al. (2024) Impact of waning immunity against SARS-CoV-2 severity exacerbated by vaccine hesitancy. PLoS Comput Biol 20(8): e1012211. https://doi.org/10.1371/ journal.pcbi.1012211

Editor: Anders Wallqvist, US Army Medical Research and Materiel Command: US Army Medical Research and Development Command, UNITED STATES OF AMERICA

Received: November 8, 2023

Accepted: May 29, 2024

Published: August 5, 2024

Copyright: © 2024 Saad-Roy et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The code to reproduce the figures is available as a Supplementary Information file.

Funding: We gratefully acknowledge funding from the Miller Institute of Basic Research in Science of UC Berkeley via a Miller Research Fellowship (to C. M.S.-R.); University of Virginia Strategic Investment Fund award number SIF160, 331 (to M.V.M. and B.L.L.); National Science Foundation **RESEARCH ARTICLE**

Impact of waning immunity against SARS-CoV-2 severity exacerbated by vaccine hesitancy

Chadi M. Saad-Roy^{1,2*}, Sinead E. Morris³, Mike Boots^{2,4}, Rachel E. Baker⁵, Bryan L. Lewis⁶, Jeremy Farrar⁷, Madhav V. Marathe^{6,8}, Andrea L. Graham⁹, Simon A. Levin⁹, Caroline E. Wagner¹⁰, C. Jessica E. Metcalf^{9,11}, Bryan T. Grenfell^{9,11}

1 Miller Institute for Basic Research in Science, University of California, Berkeley, California, United States of America, 2 Department of Integrative Biology, University of California, Berkeley, California, United States of America, 3 Department of Pathology and Cell Biology, Columbia University Medical Center, Columbia University, New York, New York, United States of America, 4 Department of Biosciences, University of Exeter, Penryn, United Kingdom, 5 Department of Epidemiology, Brown School of Public Health, Brown University, Providence, Rhode Island, University of Virginia, Charlottesville, Virginia, United States of America, 7 The Wellcome Trust, London, United Kingdom, 8 Department of Computer Science, University of Virginia, Charlottesville, Virginia, United States of America, 9 Department of Ecology and Evolutionary Biology, Princeton University, Montreal, Canada, 11 School of Public and International Affairs, Princeton University, Princeton, New Jersey, United States of America

* csaadroy@berkeley.edu

Abstract

The SARS-CoV-2 pandemic has generated a considerable number of infections and associated morbidity and mortality across the world. Recovery from these infections, combined with the onset of large-scale vaccination, have led to rapidly-changing population-level immunological landscapes. In turn, these complexities have highlighted a number of important unknowns related to the breadth and strength of immunity following recovery or vaccination. Using simple mathematical models, we investigate the medium-term impacts of waning immunity against severe disease on immuno-epidemiological dynamics. We find that uncertainties in the duration of severity-blocking immunity (imparted by either infection or vaccination) can lead to a large range of medium-term population-level outcomes (i.e. infection characteristics and immune landscapes). Furthermore, we show that epidemiological dynamics are sensitive to the strength and duration of underlying host immune responses; this implies that determining infection levels from hospitalizations requires accurate estimates of these immune parameters. More durable vaccines both reduce these uncertainties and alleviate the burden of SARS-CoV-2 in pessimistic outcomes. However, heterogeneity in vaccine uptake drastically changes immune landscapes toward larger fractions of individuals with waned severity-blocking immunity. In particular, if hesitancy is substantial, more robust vaccines have almost no effects on population-level immunoepidemiology, even if vaccination rates are compensatorily high among vaccine-adopters. This pessimistic scenario for vaccination heterogeneity arises because those few individuals that are vaccine-adopters are so readily re-vaccinated that the duration of vaccinal immunity has no appreciable consequences on their immune status. Furthermore, we find that this effect is heightened if vaccine-hesitants have increased transmissibility (e.g. due to riskier

(NSF) Grants OAC-1916805 (CINES) (to M.V.M.), CNS-2027908 (to S.A.L) and NSF-DEB-2011109 (to M.B.); NSF Expeditions in Computing Grant CCF-1918656 (to M.V.M., B.L.L., and S.A.L.), CCF-1917819 (to M.V.M. and B.L.L.); NSF RAPID 2142997 (to M.V.M. and B.L.L.); VDH Contract UVABIO610-GY23 (to M.V.M. and B.L.L.); Centers for Disease Control and Prevention (CDC) through Pathogen Genomics356 Centers of Excellence network (PGCoE) grant 6NU50CK000555-03-01 (to M.V.M and B.L.L); the Defense Threat Reduction Agency (DTRA)357 contract HDTRA120F0017 (to M.V.M and B.L.L); the James S. McDonnell Foundation 21st Century Science Initiative Collaborative Award in Understanding Dynamic and Multi-scale Systems (to S.A.L.); the C3.ai Digital Transformation Institute and Microsoft Corporation (to S.A.L.); a gift from Google, LLC (to S.A.L.); Flu Lab (to B.T.G.); and Princeton Catalysis Initiative (to B.T.G.). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

behavior). Overall, our results illustrate the necessity to characterize both transmissionblocking and severity-blocking immune time scales. Our findings also underline the importance of developing robust next-generation vaccines with equitable mass vaccine deployment.

Author summary

While the SARS-CoV-2 outbreak continues, the deployment of vaccines in many regions has blunted the severity of SARS-CoV-2 infections and decreased hospitalizations. However, the medium-term impacts of the duration of severity-blocking immunity, and its potential interactions with heterogeneous vaccine uptake (*e.g.* from vaccine hesitancy) or more robust vaccines, remain unknown. To titrate these effects, we use immuno-epidemiological models to examine potential future scenarios. We find that sufficient vaccine hesitancy (and correspondingly higher vaccination rates among adopters) can rapidly increase the fraction of individuals infected after waned severityblocking immunity even when robust vaccines are deployed. This result underlines that pharmaceutical developments for broadly protective vaccines should be combined with campaigns to increase vaccine uptake globally. We also show that this fraction is highly dependent on underlying immune uncertainties, which illustrates the importance of accurately measuring immune parameters for proper prediction based on hospitalization data.

Introduction

The severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) pandemic is a public health emergency that has had a dramatic impact across the world. In turn, it has generated a mass of epidemiological data and led to large modelling efforts [1]. Initially, guided by data analyses, a number of jurisdictions successfully implemented a range of control measures to decrease transmission, prevent a surge in infections, and decrease the burden on healthcare systems (e.g. see [2] for a retrospective analysis). In parallel, research into pharmaceutical measures (such as vaccination or therapeutics) began, with hopes to eventually control SARS-CoV-2 transmission via vaccination. Notably, with high enough coverage, transmission-blocking vaccines (i.e. that elicit immunity against infection) could lead to effective control and local elimination [3-6]. However, while the development of safe vaccines was successful (e.g. [7-9]), the susceptibility of vaccinated individuals to breakthrough infection relatively soon after vaccination (e.g. even within weeks [10]) in conjunction with the emergence of immune-escape variants (e.g. [11]) indicates that local elimination is not possible with the current generation and partial uptake of vaccines. Since the deployment of these vaccines, many jurisdictions have changed their approach for SARS-CoV-2 management to focus on mitigation against severe infections via vaccination.

Since the onset of the pandemic, a number of important gaps in our understanding of SARS-CoV-2 epidemiology have been addressed by models [1]. For example, future transmission dynamics were illuminated in an early landmark paper by Kissler et al. [12]; the role of climate and susceptibility on pandemic dynamics was investigated by Baker et al. [13, 14]; Lavine et al. [15] clarified the path to endemicity and the role of age structure; and others examined the role of novel variants [16–18]. In our previous work, we have investigated many SARS-

CoV-2 immuno-epidemiological uncertainties from a qualitative perspective. First, we used and extended a simple SIR(S) model (see [19]) to show that the relative susceptibility to infection after waning of total transmission-blocking immunity ε (so that $\varepsilon = 0$ and $\varepsilon = 1$ reduce to the SIR and SIRS models, respectively, and thus ε is a proxy for the "strength of immunity") is a key determinant of post-pandemic trajectories [6]. We then extended this framework to incorporate two-dose vaccines [20], investigate the potential effects of vaccine nationalism [21], and examine the impact of accumulating immunity on the potential future burden of chronic disease [22].

However, a number of key immuno-epidemiological questions remain. At the heart of these are uncertainties in waning immunity against severe disease (i.e. 'severity-blocking immunity'), and the ensuing potential outcomes in the medium term. In particular, from a public health standpoint, determining the likelihood, timing, and magnitude of the next surge in severe disease is crucial. Furthermore, many regions now rely on hospitalizations to monitor infection levels (especially with the pause of the ONS COVID-19 infection survey study in the UK); waning severity-blocking immunity could have an important effect on these dynamics, e.g. variations in the fraction of infections that require hospitalization at a given time, and thus crucially affect subsequent inferences. Additionally, since we have determined that the strength of immunity is a central parameter that shapes medium-term immuno-epidemiological dynamics [6], another outstanding unknown is the potential interplay between this parameter and the duration of severity-blocking immunity. Finally, given important developments toward mucosal vaccines [23-25], a major question is to determine the impacts that such vaccines with long-lasting transmission-blocking protection could have on potential outcomes and their uncertainties. For example, reducing these uncertainties may be important for robust estimates of infection levels and epidemic dynamics.

In this paper, we extend previous modelling efforts [6] to include a timescale of waning immunity against severe disease (Fig 1A). We begin with a characterization of the interplay between the strength of immunity, average duration of severity-blocking immunity, and vaccination rate, and their respective (and combined) impacts on infection levels in individuals with waned severity-blocking immunity. We then investigate the impact of vaccine characteristics on these dynamics, and we examine potential synoptic immuno-epidemiological land-scapes. Finally, we extend our model to include heterogeneities in vaccination that are driven by unequal access or hesitancy. While we cast our results in terms of vaccine hesitancy for simplicity, our findings are broadly applicable for any setting with heterogeneous uptake in vaccination.

Model framework

We extend the model of [6]. As in [6], S_P denotes the fraction of fully susceptible individuals, I_P and I_S denotes the fraction of individuals with primary and secondary infections (the latter which have relative transmissibility α), respectively, R denotes the fraction of individuals that have recovered and are fully immune, S_S denotes the fraction of individuals with waned transmission-blocking immunity and who have relative susceptibility ε to infection, and V denotes the fraction of individuals that have been vaccinated and are fully immune. Furthermore, again as in [6] μ is the birth/death rate, γ is the recovery rate, δ and δ_V are the rates of waning of natural and vaccinal transmission-blocking immunity, respectively, and β is the transmission rate (assumed to be time-dependent [13] as in previous work [6]). We additionally denote S_w as the fraction of individuals with waned severity-blocking immunity. We denote δ_{sev} and $\delta_{V,sev}$ as the rates of waning from S_S to S_w and from S_V to S_w , respectively.





The equations are as follows:

$$\frac{dS_p}{dt} = \mu - \beta(t)S_p(I_p + \alpha I_s + \alpha I_w) - \mu S_p - s_{vax}vS_p,$$
(1a)

$$\frac{dI_P}{dt} = \beta(t)S_P(I_P + \alpha I_S + \alpha I_w) - (\gamma + \mu)I_P,$$
(1b)

$$\frac{dR}{dt} = \gamma (I_P + I_S + I_w) - (\delta + \mu)R, \qquad (1c)$$

$$\frac{dS_s}{dt} = \delta R - \varepsilon \beta(t) S_s(I_p + \alpha I_s + \alpha I_w) - \mu S_s - s_{vax} v S_s - \delta_{sev} S_s,$$
(1d)

$$\frac{dI_s}{dt} = \varepsilon \beta(t) S_s(I_p + \alpha I_s + \alpha I_w) + \varepsilon \beta(t) S_V(I_p + \alpha I_s + \alpha I_w) - (\gamma + \mu) I_s,$$
(1e)

$$\frac{dV}{dt} = s_{\text{vax}} v (S_p + S_s + S_w + S_v) - (\mu + \delta_v) V, \qquad (1f)$$

$$\frac{dS_{V}}{dt} = \delta_{V}V - \varepsilon\beta(t)S_{V}(I_{P} + \alpha I_{S} + \alpha I_{w}) - (\delta_{V,sev} + \mu)S_{V} - s_{vax}vS_{V},$$
(1g)

$$\frac{dS_{w}}{dt} = \delta_{V,\text{sev}}S_{V} + \delta_{\text{sev}}S_{S} - \varepsilon\beta(t)S_{w}(I_{P} + \alpha I_{S} + \alpha I_{w}) - \mu S_{w} - s_{\text{vax}}vS_{w}, \tag{1h}$$

$$\frac{dI_w}{dt} = \varepsilon \beta(t) S_w (I_P + \alpha I_S + \alpha I_w) - (\mu + \gamma) I_w.$$
(1i)

Note that we set $\alpha = 1$ throughout and focus on ε (see [6, <u>20</u>–<u>22</u>]).

Thus, after recovery or vaccination, individuals have a period of "complete" immunity (R or V, respectively), after which they wane into partially susceptible classes (S_S and S_V , respectively), where their relative susceptibility to infection is ε . Beyond these classes, individuals eventually wane to S_w (at rates δ_{sev} and $\delta_{V,sev}$, respectively), which denotes individuals with waned severity-blocking immunity. In this class, the relative susceptibility to infection is still ε , but individuals enter a different infectious class, I_w , if they are infected after such waning (see Fig 1B for flow diagram). Thus, individuals have severity-blocking immunity while they are in R and S_S , or in V and S_V (with average durations $\frac{1}{\delta} + \frac{1}{\delta_{w}}$ and $\frac{1}{\delta_V} + \frac{1}{\delta_{W}\omega}$, respectively).

To focus on clinical severity-blocking immunity, note that we assume that the relative susceptibility of individuals in I_w and I_S are the same (and that the relative transmissibility $\alpha = 1$ in I_w and I_S is also identical). Because of this assumption, our model interpolates between a range of immune landscapes and population-level burden, while having identical transmission dynamics across scenarios (schematically depicted in Fig 1C). In one extreme case, if there is no (or very slow) waning of severity-blocking immunity, no individuals enter the compartments with waned severity-blocking immunity. On the other hand, if there is very fast waning of severity-blocking immunity, all individuals enter S_w almost immediately after complete transmission-blocking immunity wanes. Additionally, since hospitalizations (in the longer term) are likely to reflect the dynamics of I_w , we also calculate the fraction of individuals that are in this class over time (*i.e.* $\frac{I_w}{I_p+I_S+I_w}$). Note that the incorporation of vaccine hesitancy in our model is described in the S1 Text, *electronic supplementary materials*. Finally, we have produced an online interactive application (at https://grenfelllab.shinyapps.io/covid19immunity/), which can be used to examine a broad set of model scenarios.

Results and discussion

Vaccination, duration of severity-blocking immunity, and dynamics of waned infections

In Fig 2, we examine the potential epidemiological dynamics that result in changes of both severity-blocking immunity duration and vaccination coverage. We use seasonal transmission rates as in previous work [6, 22], and assume the same simple nonpharmaceutical intervention



Fig 2. Dynamics of different durations of natural and vaccinal severity protection, with variable vaccination rates, for different strengths of immunity. (*A*), (*B*), (*C*), and (*D*) have vaccination rates v = 0.0025 per week, v = 0.01 per week, v = 0.02 per week, and v = 0.04 per week, respectively. In all panels, we assume that $\frac{1}{\delta} = 0.25$ years and that $\frac{1}{\delta_v} = 0.33$ years. For each column, we assume that the duration of severity-blocking immunity imparted from vaccination or infection is the same and is equal to the columnar label ℓ_c , i.e. $\frac{1}{\delta} + \frac{1}{\delta_{wev}} = \ell_c$ and $\frac{1}{\delta_V} + \frac{1}{\delta_{vase}} = \ell_c$. Thus, $\frac{1}{\delta_{wev}} = (\ell_c - 0.25)$ years and $\frac{1}{\delta_{Vasev}} = (\ell_c - 0.33)$ years. In each panel, the *top*, *middle*, and *bottom* rows depict the fraction of individuals in I_w the fraction of infections that are in I_w (i.e. $\frac{1}{k_{vase}}$), where $I_{total} = I_P + I_S + I_w$), and the relative change in $f_{\varepsilon}(t) = \frac{I_w(t)}{I_{total}(t)}$ for each ε compared to $\varepsilon = 1$, *i.e.* $\frac{f_{\varepsilon}(t) - f_{\varepsilon}(t)}{f_{\varepsilon}(t)}$, respectively (for weeks when $f_1(t) > 0$). Other parameters are $\gamma = \frac{7}{5}$ week⁻¹ and $\mu = 0.02$ years⁻¹, as in previous work [6, 20–22]. The initial conditions here and throughout are a fraction 10^{-9} of individuals with primary infection (I_P) and the remainder fully susceptible (S_P), which is as in previous work with the simpler model [6].

https://doi.org/10.1371/journal.pcbi.1012211.g002

settings as in [22]. (For a specific expression for $\beta(t)$ in the absence of nonpharmaceutical intervention, see [6], which uses values derived by [13].) For each vaccination rate (*i.e.* each panel of Fig 2), we plot the fraction of individuals that are infected with waned severity-blocking immunity (*i.e.* I_w) (top row), the fraction of all infections that I_w represents (*i.e.* $\frac{I_w}{I_{ww}}$) (middle row), and the relative change in this latter fraction compared to complete susceptibility to reinfection after transmission-blocking immunity wanes (*i.e.* $\varepsilon = 1$ giving the SIRS model, see the caption of Fig 2 for mathematical details) (bottom row). For all panels of Fig 2, we assume the same average duration of transmission-blocking vaccinal (0.33 years) and natural (0.25 years) immunity. Note that while we take these to be relatively short (as in [22]) since infection can happen relatively rapidly after recovery or vaccination (e.g. for infection after vaccination see [10]), the effects of longer durations can be examined thoroughly with our companion interactive online application (at https://grenfelllab.shinyapps.io/covid19immunity/). Furthermore, in each panel of Fig 2, we assume that the durations of vaccinal and natural severity-blocking immunity are the same across a column (identified by the columnar label). Additionally, note that since the vaccination rate is fixed within each panel, the transmission dynamics across columns of each panel for a fixed value of ε are identical. However, because of the additional immune time scale, the underlying immunity landscapes change, leading to differences in infection characteristics and potential burden.

Intuitively, as the duration of severity-blocking immunity increases, the fraction of individuals with infections after severity-blocking immunity has waned decreases (compare left to right plots of the top rows of Fig 2A–2D). Similarly, driven by more frequent boosting of immunity, higher vaccination rates result in further decreases (compare top rows of Fig 2A– 2D). Furthermore, a lower relative susceptibility to reinfection (*i.e.* stronger immunity) initially leads to a smaller fraction of infections with waned severity (*middle rows*, Fig 2A–2D). However, especially for lower vaccination rates, an increase in the strength of immunity (lower ε) can potentially lead to larger fractions of infections that are in I_w (*middle* and *bottom rows*, Fig 2A). Additionally, intermediate values of ε can lead to larger peaks (and deeper troughs) in I_w (*top row*, Fig 2A). Interestingly, these results are partially reminiscent of the findings of [6], where, in some scenarios, stronger immunity can lead to a bigger (and delayed) second peak in infections. Finally, very high vaccination coverage (Fig 2) dampens these effects because of more frequent gains in immunity.

Overall, these results illustrate an additional potential complication associated with predicting the number of total infections based on hospitalizations alone, in addition to a wide variety of known difficulties. To reduce this particular complexity, such predictions would likely necessitate robust parameter estimates for the strength of immunity and the duration of severity-blocking immunity. These could be obtained from large immuno-epidemiological cohort studies, echoing previous calls for such monitoring [26–29].

Future vaccine refinements

So far, we have assumed that the period of complete immunity imparted by vaccination is transient, with relatively high susceptibility after waning. While this reflects current settings with existing vaccines (*e.g.* in part due to circulating immune-escape variants), pan-coronavirus and pan-sarbecovirus vaccines [30] are in development. Additionally, there have been recent landmark advances in the development of mucosal vaccines [23, 24], which would likely be able to more successfully block transmission. Furthermore, it seems that such a mucosal vaccine could generate immunity across sarbecoviruses [23], and thus potentially generate broad immune responses to novel SARS-CoV-2 variants. In Fig.3, we examine the impact of a more durable transmission-blocking vaccine on severity dynamics, for intermediate (1% per week)



Fig 3. Impacts of longer transmission-blocking vaccines on severity dynamics. In (*A*) and (*B*), the vaccination rates are 0.01 per week and 0.02 per week, respectively. In both panels, the top row denotes the total fraction $I_{\text{total}} = I_P + I_S + I_w$ of individuals that are infected. The second to fourth rows are as in the rows of each panel of Fig 2 (see caption of Fig 2 for definitions). Across both panels, we assume that the duration of vaccinal transmission-blocking immunity is 90% of the duration of severity-blocking immunity (the columnar label), and that transmission-blocking and severity-blocking immunity after infection last 0.25 years and 1.5 years, respectively (*i.e.* $\frac{1}{\delta} = 0.25$ years and $\frac{1}{\delta} + \frac{1}{\delta_m} = 1.5$ years).

and high (2% per week) vaccination rates (panels *A* and *B*, respectively). To allow for appropriate comparisons within and across panels, we assume that the duration of vaccinal severityblocking immunity is conserved within a column (indicated by the columnar label) within each panel. Furthermore, we assume that for 90% of that duration, vaccinal immunity also fully blocks transmission. (Note that this contrasts with Fig 2, where the average complete vaccinal immunity was assumed to be 0.33 years.) Finally, we take a moderately optimistic assumption and assume that severity-blocking immunity after infection lasts on average 1.5 years (in Fig 2, this value was varied concurrently with that imparted following vaccination). Finally, in each panel, the top row denotes the total infections over time, and the bottom three rows are as in those of each panel of Fig 2.

Even for an intermediate vaccination rate, a more durable vaccine leads to fewer total infections (Fig 3, *top row*, compare left to right plots) and fewer infections after severity-blocking immunity has waned (*second row*). Furthermore, with increases in durability, the fraction $\frac{I_w}{I_{total}}$ depends increasingly less on the strength of immunity (*third and bottom rows*, Fig 3, compare left to right plots). In a different setting, this decrease in dependence on ε is akin to that observed in Fig 2 for very high vaccination rates. Intuitively, a more durable vaccine is analogous to very high vaccination rates (*i.e.* more frequent boosting) with a less durable vaccine. Finally, a high vaccination rate further accentuates the effects of a durable vaccine on epidemiological dynamics (compare Fig 3A and 3B). Thus, the development and deployment of a durable vaccine, combined with a high vaccination rate, can substantially reduce uncertainties in outcomes.

Immuno-epidemiological outlooks

So far, we have examined changes in severity dynamics via total and relative infection levels across a range of settings for the strength of immunity and durations of both severity-blocking and transmission-blocking immunity. In Fig 4, we summarize synoptic medium-term immuno-epidemiological scenarios based on optimistic or pessimistic assumptions on severity-blocking immunity, different vaccination rates, and changes in durability of vaccines. For each scenario, we present time series of infections after severity-blocking immunity has waned and of the fraction of infections that these consist of. Below, we illustrate immunity and infection phenotypes over time. Note that at the bottom of each such area plot are the three infection types $(I_P, I_S, \text{ and } I_w)$, and thus the total fraction of individuals infected is immediately seen visually. While we had previously assumed in Fig 2 (for existing vaccines) that the durations of vaccinal and natural severity-blocking immunity were equal, we now relax this assumption in optimistic scenarios for waning of severity-blocking immunity (*i.e.* second column of Fig 4) and assume in those settings that the average duration of vaccinal severity-blocking immunity is (optimistically) slightly longer than that of natural severity-blocking immunity (2 years instead of 1.5 years). For a more durable vaccine, we assume that transmission-blocking immunity lasts on average 1.33 years, and that severity-blocking immunity lasts on average either 1.5 years (in the more pessimistic scenario with faster waning) or 3 years (if waning is optimistically slower).

With a low vaccination rate, vaccinal characteristics have limited impact on immunoepidemiological dynamics (compare left two plots and right two plots, *top row*, Fig 4). However, the relative time scale of waning severity-blocking immunity drastically alters the immune landscape (*top row*, Fig 4). With an intermediate vaccination rate, a durable vaccine has important dynamical impacts (as also seen in [20, 22]) (*middle row*, Fig 4). Intermediate vaccination rates also partially modulate pessimistic outcomes if severity wanes rapidly; this is further emphasized if vaccination is increased further (Fig 4). However, if severity-blocking immunity wanes rapidly and a vaccine does not provide long-lasting transmission-blocking protection, then the buildup of susceptibles with waned severity-blocking immunity remains substantial irrespective of vaccination rates (compare leftmost plots of each row, Fig 4). Thus, to decrease this accumulation, a high vaccination rate with a more durable vaccine is necessary.



low vaccination rate (0.25% per week)

Fig 4. Synoptic landscapes of severity-blocking immunity. The *top*, *middle* and *bottom* rows have vaccination rates 0.0025, 0.01, and 0.02 per week, respectively. The *leftmost* two columns illustrate scenarios with a less durable vaccine, *i.e.* $\frac{1}{\delta_V} = 0.33$ years, whereas the *rightmost* two columns represent scenarios with a more durable vaccine, *i.e.* $\frac{1}{\delta_V} = 1.33$ years. The *first* and *third* columns assume faster waning of severity-blocking immunity, with the *first* column having $\frac{1}{\delta_V} + \frac{1}{\delta_{V,sev}} = \frac{1}{\delta} + \frac{1}{\delta_{sev}} = 0.5$ years and the *third* column having $\frac{1}{\delta_V} + \frac{1}{\delta_{V,sev}} = 1.5$ years (since the vaccine is more durable) and $\frac{1}{\delta} + \frac{1}{\delta_{sev}} = 0.5$ years. On the other hand, the *second* and *fourth* columns assume slower waning of severity-blocking immunity, with the *second* column having $\frac{1}{\delta} + \frac{1}{\delta_{sev}} = 1.5$ years and $\frac{1}{\delta_V} + \frac{1}{\delta_{V,sev}} = 2$ years, and the *fourth* column having $\frac{1}{\delta} + \frac{1}{\delta_{sev}} = 1.5$ years and $\frac{1}{\delta_V} + \frac{1}{\delta_{V,sev}} = 2$ years, and the *fourth* column having $\frac{1}{\delta} + \frac{1}{\delta_{sev}} = 1.5$ years. In each panel, the left and right axes of the top plot are I_w and the fraction $\frac{I_w}{I_{p+I_S+I_w}}$ respectively, and the area plot colours correspond to the compartments in Fig 1B. In all panels, $\varepsilon = 0.8$. All other parameters are as in Fig 2 and 3, and the colours in the area plots are as in Fig 1B.

Heterogeneities in vaccination coverage

Current vaccination rates are very variable globally and at local scales, both due to inequity in supply and hesitancy. For example, uptake of bivalent booster doses in the United States has been low, even among those who received the initial vaccines [31]. In a specific region, inequity in supply can arise from a number of issues, including due to vaccine nationalism by other regions [21], and vaccine hesitancy can emerge from underlying behavioural drivers [32, 33]. As shown and discussed in previous work, these heterogeneities in vaccination can have important immuno-epidemiological impacts on the medium- and long-term dynamics of SARS-CoV-2 (e.g. [6, 34]).

To investigate the potential consequences of vaccination heterogeneity on medium-term immune landscapes and burden due to infections after waning of severity-blocking immunity, we consider a simple extension of our basic framework with the addition of a group whose individuals never receive vaccinations, but otherwise mix homogeneously with individuals that are in the vaccine-adopter group (see <u>S1 Text</u>, *electronic supplementary materials*, for model equations). In Fig 5, we illustrate the medium-term outcomes for a range of vaccinehesitant group sizes (*rows*), for different scenarios of severity-blocking immunity and vaccine characteristics (*columns*). In all these panels, we assume that the average vaccination rate is 2% per week (*i.e.* $vN_1 = 0.02$, where $N_1 = 1 - N_2$ is the fraction of individuals that are vaccineadopters and N_2 is the fraction of individuals that are never vaccinated), which corresponds to a 'high' vaccination scenario in the homogeneous setting of Fig 4.

Direct comparisons between Fig 4 (bottom row) and the rows of Fig 5 reveal that the homogeneous vaccination assumption is an optimistic upper bound. In particular, vaccination heterogeneity increases the fraction of infections after severity-blocking immunity has waned, and can even lead to recurrent outbreaks if there are very few individuals that are receiving vaccinations. Intuitively, these observations emerge because vaccine-adopters are a heavily vaccinated group of individuals who are re-vaccinated often. On the other hand, individuals with waned immunity that are never vaccinated can only regain immunity via infection. Thus, even in an optimistic scenario where there is slower waning of severityblocking immunity, a large fraction of these never-vaccinated individuals have waned severity-blocking immunity. Furthermore, if there is a substantial fraction of individuals that are not receiving vaccination, a more durable vaccine has almost no immuno-epidemiological effect on the population-level dynamics (compare left two columns with right two columns, respectively, of Fig 5, bottom row). Finally, if the average vaccination rate decreases, the impact of the resulting vaccine heterogeneity on immuno-epidemiological outcomes is slightly attenuated (e.g., see S1 Fig, electronic supplementary materials where the average rate is 0.01 per week).

In Fig 6, we examine the cumulative number of infections in I_w that occur after the onset of vaccination up to year 5 (relative to population size), as a function of the fraction of individuals that are vaccine-hesitant. As in Fig 5, we assume that the average vaccination rate is 2% per week. Across scenarios, an increase in vaccine hesitancy leads to a greater number of infections in I_w . This effect is further magnified if severity-blocking immunity wanes rapidly (compare *left* with *right* panels of Fig 6, or if vaccine-hesitants have a higher transmissibility. As illustrated in Fig 5, vaccine hesitancy counters the deployment of a more durable vaccine (*bottom* panels, Fig 6). In particular, sufficient vaccine hesitancy can lead to a sharp increase in the cumulative number of infections in I_w (*bottom left* panel, Fig 6). Thus, if the fraction of individuals that are vaccine-hesitant is below this threshold, there are very few infections with waned severity-blocking immunity. However, if the fraction of individuals that are vaccine-hesitant increases further beyond this threshold, the cumulative number is substantially



40% are vaccine-hesitant

Fig 5. Synoptic landscapes with vaccine heterogeneities, caused by either unequal access or hesitancy. We assume a 2% weekly vaccination rate (*c.f. bottom row*, Fig 4), and keep the average vaccination rate constant across each row so that the vaccination rate among vaccine-adopters is v, where $vN_1 = 0.02$ ($N_1 = 1 - N_2$ is the fraction of vaccine adopters, and N_2 is the fraction of individuals that are vaccine-hesitant). The columnar scenarios are as in those of Fig 4.

https://doi.org/10.1371/journal.pcbi.1012211.g005

increased. If vaccine-hesitants have a higher transmissibility, we find that this threshold occurs for a much smaller level of vaccine hesitancy (*bottom left panel*, Fig 6), which can impact immuno-epidemiological dynamics (see <u>S2 Fig</u>, *electronic supplementary materials* for an example). Overall, these results illustrate that, even with corresponding adjustments to



Fig 6. Cumulative infections with waned severity-blocking immunity after the onset of vaccination up to year 5 as a function of the fraction of individuals that are vaccine-hesitant. The *top left, top right, bottom left,* and *bottom right* panel depict the same scenarios as the first, second, third, and fourth columns of Figs 4 and 5, respectively. As in Fig 5, the average vaccination rate is constant. In each panel, the different lines denote different relative transmissibility values for vaccine hesitants.

vaccination rates among adopters, vaccine hesitancy can substantially hinder the epidemiological benefits of more durable vaccines.

Caveats and future directions

To distill the impacts of severity-blocking immunity on potential medium-term outcomes, we have made a number of simplifying assumptions in our modelling framework that should be relaxed in future work. First, we have ignored vaccine dosing regimes (see *e.g.* [20]) and assumed that individuals get vaccinated at some rate, with each subsequent vaccine they obtain giving rise to similar vaccinal immunity. In reality, multiple doses can lead to more robust immunity, and incorporating explicit vaccine doses in our model could reveal subsequent

impacts. Relatedly, we have ignored the potential accumulation of immunity (whether transmission-blocking or against severity) after multiple exposures. Combining this refined model with that of [22] could elucidate any intricacies that may emerge due to the interaction between accumulating immunity and waning severity-blocking immunity. We have also ignored the impact of time-dependent variable vaccination rates, and incorporating this with specific vaccination data for various regions would be valuable. Notwithstanding these complexities, we have shown that the qualitative impact of hesitancy on vaccine performance is robust to underlying assumptions. This underlines the importance of more refined and granular models for dynamics of hesitancy in future work.

While we have examined heterogeneities in vaccination, we have ignored various other heterogeneities, *e.g.* in transmission [35, 36], or due to age [15] or vulnerabilities [37]. Exploring these further, and their confluence with vaccination heterogeneities and severity-blocking immunity, is an important future direction. In particular, we have assumed that interactions between individuals that are vaccine-hesitant and those that adopt vaccines are homogeneous. In reality, however, interactions within a group could be more likely than between groups. These features could enhance transmission potential and reduce the likelihood of control via vaccination (see *e.g.* [6] for a simple consideration of this), and examining the interplay between these effects and waning severity-blocking immunity is an important are of future research.

Relatedly, we have ignored the dynamics of human behavior [38], especially regarding adherence to nonpharmaceutical interventions (*e.g.* [39, 40]) or vaccination [41, 42]. However, the potential feedbacks between these social and epidemiological dynamics could shape immuno-epidemiological trajectories. Thus, incorporating these features into an epidemiological-behavioral model with severity-blocking immunity would be particularly fruitful.

We have also omitted individual variations in viral loads and immune kinetics. In particular, it would be particularly insightful to formulate cross-scale models that couple our framework with within-host dynamics. Coupled with a model for viral evolution, such a framework could potentially aid in understanding viral phylodynamics of SARS-CoV-2 (see [29, 43, 44]). Relatedly, we have ignored the dynamics of Long COVID, and exploring the connections between this, severity-blocking immunity, and potential medium-term chronic burden is a salient avenue for future work. Overall, understanding the impacts of these various heterogeneities will require complex models with comprehensive data (*e.g.* from large cohort studies [29]).

In line with previous work (see *e.g.* [6, 20–22, 45]) we have assumed that NPIs decrease transmission by a fixed value for fixed periods of time. However, NPIs are often implemented dynamically. Furthermore, in the absence of mandated NPIs, individuals may still choose to adhere to certain interventions (*e.g.* mask-wearing, social distancing). Incorporating the underlying social dynamics that then determine NPI adherence in such a setting may be important [39]. Thus, a potentially fruitful future avenue would be to couple our simple immuno-epidemiological models with more realistic formulations of NPI adherence, calibrated to particular regions of interest.

The impacts of vaccination, transmission-blocking and severity-blocking immunity, vaccine hesitancy, varying periods of NPIs, and climatic effects on transmission can be further explored using the interactive online application at <u>https://grenfelllab.shinyapps.io/</u> covid19immunity/.

Conclusion

As the SARS-CoV-2 outbreak continues to progress, testing and monitoring of infections has been widely relaxed and the public health emergency of international concern (PHEIC) has

ended, but transmission remains high. In parallel, while the mass of data accumulated so far has improved our understanding of host immune responses following infection or vaccination, a number of uncertainties remain, especially in the duration of immunity against severe disease and in the relative susceptibility to reinfection after waning of transmission-blocking immunity.

Our simple models reveal that a large range of outcomes can emerge from uncertainties in both the duration of severity-blocking immunity and the strength of immunity, and from the confluence of these two parameters. In particular, our findings emphasize that the strength of immunity shapes immuno-epidemiological dynamics at multiple resolutions, and that the duration of severity-blocking immunity has a major effect on population-level immune land-scapes and potential burdens. Thus, to properly infer infection dynamics from hospitalization data, accurate estimates of both these parameters are needed, which could be accomplished via future cohort studies monitoring immuno-epidemiology [29] and a Global Immunological Observatory [26–28].

Finally, we have also shown that high vaccination rates, in combination with a more durable vaccine, can alleviate pessimistic outcomes for both the buildup of susceptible individuals with waned severity-blocking immunity and for the level of infections with waned severity-blocking immunity. Our results also illustrate the importance of broad vaccination coverage, echoing previous findings that argued for equities in vaccination access [20–22, 34]. In particular, we find that ignoring the specter of vaccine hesitancy in regions awash with vaccines can, at the population-level, essentially counteract important pharmaceutical developments to improve vaccine breadth and strength. Since we have shown that this result is generally robust to model assumptions on underlying uncertainties of severity-blocking immunity, our work underlines the need to identify and understand the behavioural drivers of vaccine hesitancy [33]. In tandem, since the impact of hesitancy is especially amplified if vaccine-hesitants have higher transmissibility, our results further stress the importance of nonpharmaceutical interventions in regions with elevated levels of hesitancy. Overall, to prevent pessimistic outcomes from waning severity-blocking immunity, increases in global vaccination rates in conjunction with the development of more robust vaccines are necessary.

Supporting information

S1 Fig. Vaccine hesitancy with a lower baseline vaccination rate. This figure is as in Fig 5, but with an average vaccination rate of 1% per week (instead of the 2% per week in Fig 5). (PDF)

S2 Fig. The impacts of increased transmissibility of vaccine hesitants on immuno-epidemiological dynamics. This figure is as in Fig 5, but with $\alpha_V = 1.2$. (PDF)

S1 Text. Supplementary information text. (PDF)

S1 File. Zip file with R code. (ZIP)

Author Contributions

Conceptualization: Chadi M. Saad-Roy, Bryan L. Lewis, Bryan T. Grenfell.

Formal analysis: Chadi M. Saad-Roy, Bryan T. Grenfell.

- Investigation: Chadi M. Saad-Roy, Sinead E. Morris, Mike Boots, Rachel E. Baker, Bryan L. Lewis, Jeremy Farrar, Madhav V. Marathe, Andrea L. Graham, Simon A. Levin, Caroline E. Wagner, C. Jessica E. Metcalf, Bryan T. Grenfell.
- Software: Chadi M. Saad-Roy, Sinead E. Morris.
- Writing original draft: Chadi M. Saad-Roy.
- Writing review & editing: Chadi M. Saad-Roy, Sinead E. Morris, Mike Boots, Rachel E. Baker, Bryan L. Lewis, Jeremy Farrar, Madhav V. Marathe, Andrea L. Graham, Simon A. Levin, Caroline E. Wagner, C. Jessica E. Metcalf, Bryan T. Grenfell.

References

- Koelle K, Martin MA, Antia R, Lopman B, Dean NE. The changing epidemiology of SARS-CoV-2. Science. 2022; 375(6585):1116–1121. https://doi.org/10.1126/science.abm4915 PMID: 35271324
- Flaxman S, Mishra S, Gandy A, Unwin HJT, Mellan TA, Coupland H, et al. Estimating the effects of non-pharmaceutical interventions on COVID-19 in Europe. Nature. 2020; 584(7820):257–261. https:// doi.org/10.1038/s41586-020-2405-7 PMID: 32512579
- 3. Anderson RM, May RM. Infectious diseases of humans: dynamics and control. Oxford, UK: Oxford University Press; 1991.
- Anderson RM, Vegvari C, Truscott J, Collyer BS. Challenges in creating herd immunity to SARS-CoV-2 infection by mass vaccination. The Lancet. 2020; 396(10263):1614–1616. <u>https://doi.org/10.1016/ S0140-6736(20)32318-7</u> PMID: 33159850
- Saad-Roy CM, Levin SA, Metcalf CJE, Grenfell BT. Trajectory of individual immunity and vaccination required for SARS-CoV-2 community immunity: a conceptual investigation. Journal of The Royal Society Interface. 2021; 18(175):20200683. https://doi.org/10.1098/rsif.2020.0683 PMID: 33530857
- Saad-Roy CM, Wagner CE, Baker RE, Morris SE, Farrar J, Graham AL, et al. Immune life history, vaccination, and the dynamics of SARS-CoV-2 over the next 5 years. Science. 2020; 370(6518):811–818. https://doi.org/10.1126/science.abd7343 PMID: 32958581
- Polack FP, Thomas SJ, Kitchin N, Absalon J, Gurtman A, Lockhart S, et al. Safety and Efficacy of the BNT162b2 mRNA Covid-19 Vaccine. New England Journal of Medicine. 2020; 383(27):2603–2615. https://doi.org/10.1056/NEJMoa2034577 PMID: 33301246
- Baden LR, El Sahly HM, Essink B, Kotloff K, Frey S, Novak R, et al. Efficacy and Safety of the mRNA-1273 SARS-CoV-2 Vaccine. New England Journal of Medicine. 2021; 384(5):403–416. https://doi.org/ 10.1056/NEJMoa2035389 PMID: 33378609
- Voysey M, Clemens SAC, Madhi SA, Weckx LY, Folegatti PM, Aley PK, et al. Safety and efficacy of the ChAdOx1 nCoV-19 vaccine (AZD1222) against SARS-CoV-2: an interim analysis of four randomised controlled trials in Brazil, South Africa, and the UK. The Lancet. 2021; 397(10269):99–111. https://doi. org/10.1016/S0140-6736(20)32661-1 PMID: 33306989
- Antonelli M, Penfold RS, Merino J, Sudre CH, Molteni E, Berry S, et al. Risk factors and disease profile of post-vaccination SARS-CoV-2 infection in UK users of the COVID Symptom Study app: a prospective, community-based, nested, case-control study. The Lancet Infectious Diseases. 2022; 22(1):43– 55. https://doi.org/10.1016/S1473-3099(21)00460-6 PMID: 34480857
- Planas D, Saunders N, Maes P, Guivel-Benhassine F, Planchais C, Buchrieser J, et al. Considerable escape of SARS-CoV-2 Omicron to antibody neutralization. Nature. 2021; p. https://doi.org/10.1038/ s41586-021-04389-z PMID: 35016199
- Kissler SM, Tedijanto C, Goldstein E, Grad YH, Lipsitch M. Projecting the transmission dynamics of SARS-CoV-2 through the postpandemic period. Science. 2020; 368(6493):860–868. https://doi.org/10. 1126/science.abb5793 PMID: 32291278
- Baker RE, Yang W, Vecchi GA, Metcalf CJE, Grenfell BT. Susceptible supply limits the role of climate in the early SARS-CoV-2 pandemic. Science. 2020; 369:315–319. <u>https://doi.org/10.1126/science.</u> abc2535 PMID: 32423996
- Baker RE, Yang W, Vecchi GA, Metcalf CJE, Grenfell BT. Assessing the influence of climate on wintertime SARS-CoV-2 outbreaks. Nature Communications. 2021; 12(1):846. <u>https://doi.org/10.1038/</u> s41467-021-20991-1 PMID: 33558479
- Lavine JS, Bjornstad ON, Antia R. Immunological characteristics govern the transition of COVID-19 to endemicity. Science. 2021; 371:741–745. https://doi.org/10.1126/science.abe6522 PMID: 33436525

- Dyson L, Hill EM, Moore S, Curran-Sebastian J, Tildesley MJ, Lythgoe KA, et al. Possible future waves of SARS-CoV-2 infection generated by variants of concern with a range of characteristics. Nature Communications. 2021; 12(1):5730. https://doi.org/10.1038/s41467-021-25915-7 PMID: 34593807
- Davies NG, Abbott S, Barnard RC, Jarvis CI, Kucharski AJ, Munday JD, et al. Estimated transmissibility and impact of SARS-CoV-2 lineage B.1.1.7 in England. Science. 2021; 372(6538):eabg3055. <u>https:// doi.org/10.1126/science.abg3055</u> PMID: 33658326
- Davies NG, Jarvis CI, van Zandvoort K, Clifford S, Sun FY, Funk S, et al. Increased mortality in community-tested cases of SARS-CoV-2 lineage B.1.1.7. Nature. 2021; 593(7858):270–274. <u>https://doi.org/ 10.1038/s41586-021-03426-1</u> PMID: 33723411
- Morris SE, Pitzer VE, Viboud C, Metcalf CJE, Bjørnstad ON, Grenfell BT. Demographic buffering: titrating the effects of birth rate and imperfect immunity on epidemic dynamics. Journal of The Royal Society Interface. 2015; 12(104):20141245. https://doi.org/10.1098/rsif.2014.1245 PMID: 25589567
- Saad-Roy CM, Morris SE, Metcalf CJE, Mina MJ, Baker RE, Farrar J, et al. Epidemiological and evolutionary considerations of SARS-CoV-2 vaccine dosing regimes. Science. 2021; 372(6540):363–370. https://doi.org/10.1126/science.abg8663 PMID: 33688062
- Wagner CE, Saad-Roy CM, Morris SE, Baker RE, Mina MJ, Farrar J, et al. Vaccine nationalism and the dynamics and control of SARS-CoV-2. Science. 2021; 373(6562):eabj7364. https://doi.org/10.1126/ science.abj7364 PMID: 34404735
- Saad-Roy CM, Morris SE, Baker RE, Farrar J, Graham AL, Levin SA, et al. Medium-term scenarios of COVID-19 as a function of immune uncertainties and chronic disease. Journal of The Royal Society Interface. 2023; 20(205):20230247. https://doi.org/10.1098/rsif.2023.0247 PMID: 37643641
- Mao T, Israelow B, Peña-Hernández MA, Suberi A, Zhou L, Luyten S, et al. Unadjuvanted intranasal spike vaccine elicits protective mucosal immunity against sarbecoviruses. Science. 2022; 378(6622): eabo2523. https://doi.org/10.1126/science.abo2523 PMID: 36302057
- Tang J, Zeng C, Cox TM, Li C, Son YM, Cheon IS, et al. Respiratory mucosal immunity against SARS-CoV-2 after mRNA vaccination. Science Immunology. 2022; 7(76):eadd4853. https://doi.org/10.1126/ sciimmunol.add4853 PMID: 35857583
- Topol EJ, Iwasaki A. Operation Nasal Vaccine–Lightning speed to counter COVID-19. Science Immunology. 2022; 7(74):eadd9947. https://doi.org/10.1126/sciimmunol.add9947 PMID: 35862488
- Metcalf CJE, Farrar J, Cutts FT, Basta NE, Graham AL, Lessler J, et al. Use of serological surveys to generate key insights into the changing global landscape of infectious disease. The Lancet. 2016; 388:728–730. https://doi.org/10.1016/S0140-6736(16)30164-7 PMID: 27059886
- Metcalf CJE, Mina MJ, Winter AK, Grenfell BT. Opportunities and challenges of a World Serum Bank— Authors' reply. The Lancet. 2017; 389:252. <u>https://doi.org/10.1016/S0140-6736(17)30054-5</u> PMID: 28118913
- Mina MJ, Metcalf CJE, McDermott AB, Douek DC, Farrar J, Grenfell BT. A Global Immunological Observatory to meet a time of pandemics. eLife. 2020; 9:e58989. <u>https://doi.org/10.7554/eLife.58989</u> PMID: 32510329
- 29. Saad-Roy CM, Metcalf CJE, Grenfell BT. Immuno-epidemiology and the predictability of viral evolution. Science. 2022; 376(6598):1161–1162. https://doi.org/10.1126/science.abn9410 PMID: 35679395
- Park YJ, Marco AD, Starr TN, Liu Z, Pinto D, Walls AC, et al. Antibody-mediated broad sarbecovirus neutralization through ACE2 molecular mimicry. Science. 2022; 375(6579):449–454. <u>https://doi.org/10.1126/science.abm8143</u> PMID: 34990214
- Lu Pj, Zhou T, Santibanez TA, Jain A, Black CL, Srivastav A, et al. COVID-19 bivalent booster vaccination coverage and intent to receive booster vaccination among adolescents and adults–United States, November–December 2022. Morbidity and Mortality Weekly Report. 2023; 72(7):190–198. <u>https://doi.org/10.15585/mmwr.mm7207a5 PMID: 36795677</u>
- Dubé E, Laberge C, Guay M, Bramadat P, Roy R, Bettinger JA. Vaccine hesitancy: an overview. Human Vaccines and Immunotherapeutics. 2013; 9:11763–1773. <u>https://doi.org/10.4161/hv.24657</u> PMID: 23584253
- Wagner CE, Prentice JA, Saad-Roy CM, Yang L, Grenfell BT, Levin SA, et al. Economic and Behavioral Influencers of Vaccination and Antimicrobial Use. Frontiers in Public Health. 2020; 8:6141113. <u>https:// doi.org/10.3389/fpubh.2020.614113</u> PMID: 33409264
- Wagner CE, Saad-Roy CM, Grenfell BT. Modelling vaccination strategies for COVID-19. Nature Reviews Immunology. 2022; 22(3):139–141. https://doi.org/10.1038/s41577-022-00687-3 PMID: 35145245
- Laxminarayan R, Wahl B, Dudala SR, Gopal K, B CM, Neelima S, et al. Epidemiology and transmission dynamics of COVID-19 in two Indian states. Science. 2020; 370(6517):691–697. <u>https://doi.org/10.1126/science.abd7672</u> PMID: 33154136

- Althouse BM, Wenger EA, Miller JC, Scarpino SV, Allard A, Hébert-Dufresne L, et al. Stochasticity and heterogeneity in the transmission dynamics of SARS-CoV-2. arXiv. 2020; p. https://doi.org/10.48550/ arXiv.2005.13689.
- Rice BL, Annapragada A, Baker RE, Bruijning M, Dotse-Gborgbortsi W, Mensah K, et al. Variation in SARS-CoV-2 outbreaks across sub-Saharan Africa. Nature Medicine. 2021; 27(3):447–453. <u>https://doi.org/10.1038/s41591-021-01234-8 PMID</u>: 33531710
- Bergstrom CT, Hanage WP. Human behavior and disease dynamics. Proceedings of the National Academy of Sciences. 2024; 121(1):e2317211120. https://doi.org/10.1073/pnas.2317211120 PMID: 38150502
- Traulsen A, Levin SA, Saad-Roy CM. Individual costs and societal benefits of interventions during the COVID-19 pandemic. Proceedings of the National Academy of Sciences. 2023; 120(24):e2303546120. https://doi.org/10.1073/pnas.2303546120 PMID: 37285394
- Saad-Roy CM, Traulsen A. Dynamics in a behavioral?epidemiological model for individual adherence to a nonpharmaceutical intervention. Proceedings of the National Academy of Sciences. 2023; 120(44): e2311584120. https://doi.org/10.1073/pnas.2311584120
- Bauch CT, Earn DJD. Vaccination and the theory of games. Proceedings of the National Academy of Sciences. 2004; 101(36):13391–13394. https://doi.org/10.1073/pnas.0403823101 PMID: 15329411
- Bauch CT. Imitation dynamics predict vaccinating behaviour. Proceedings of the Royal Society B: Biological Sciences. 2005; 272(1573):1669–1675. <u>https://doi.org/10.1098/rspb.2005.3153</u> PMID: 16087421
- Grenfell BT, Pybus OG, Gog JR, Wood JLN, Daly JM, Mumford JA, et al. Unifying the Epidemiological and Evolutionary Dynamics of Pathogens. Science. 2004; 303(5656):327–332. https://doi.org/10.1126/ science.1090727 PMID: 14726583
- Volz EM, Koelle K, Bedford T. Viral Phylodynamics. PLOS Computational Biology. 2013; 9(3): e1002947. https://doi.org/10.1371/journal.pcbi.1002947 PMID: 23555203
- 45. Baker RE, Saad-Roy CM, Park SW, Farrar J, Metcalf CJE, Grenfell BT. Long-term benefits of nonpharmaceutical interventions for endemic infections are shaped by respiratory pathogen dynamics. Proceedings of the National Academy of Sciences. 2022; 119(49):e2208895119. https://doi.org/10.1073/ pnas.2208895119 PMID: 36445971